Author's response to reviews

Title: The Effect of Continuous Ultrasound on Chronic Low Back Pain: Protocol of a Randomized Controlled Trial

Authors:

Safoora Ebadi (s_ebadi@razi.tums.ac.ir)
Noureddin Nakhostin Ansari (nakhostin@sina.tums.ac.ir)
Nicholas Henschke (nhenschke@georgeinstitute.org.au)
Soofia Naghdi (naghdi@sina.tums.ac.ir)
Maurits W van Tulder (maurits.van.tulder@falw.vu.nl)

Version: 3 Date: 9 March 2011

Author's response to reviews:

To the editor:
The effect of continuous ultrasound on chronic low back pain: protocol of a randomized controlled trial
I would like you to know that
• the trial registry number is added at the end of the abstract.
• The title is rewritten in the correct format
• The highlighting are removed